Author's response to reviews

Title: Nodular osteochondrogenic activity in soft tissue surrounding osteoma in Neurogenic Para Osteo-Arthropathy: Morphological and immunohistochemical study.

Authors:

Tayebeh Youssefian (tayebeh.youssefian@wanadoo.fr)
Rosa Sapena (rosa.sapena@sympathos.fr)
Robert Y Carlier (robert.carlier@rpc.ap-hop-paris.fr)
Christophe Bos (christophe.bos@sympathos.fr)
Anne Cormier (anne.cormier@sympathos.fr)
Philippe Denormandie (philippe.denormandie@rpc.ap-hop-paris.fr)
Pierre Denys (pierre.denys@rpc.ap-hop-paris.fr)
Michel A Bandelier (michel.bandelier@sympathos.fr)

Version: 3 Date: 11 October 2004

Author's response to reviews: see over
Paris, 04 December 2003

The Editorial office of
The Journal of BMC
“Musculoskeletal disorders”

Dear Editor,

Please find enclosed the manuscript entitled « Nodular osteochondrogenic activity in soft tissue surrounding osteoma in Neurologic Para Osteo-Arthropathy: Morphological and immunohistochemical study.».

I would like to submit it for publication in the Journal of BMC “Musculoskeletal disorders”.

Hoping it will seem to you of interest,

Sincerely yours.

Tayebeh Youssefian